Author’s response to reviews

Title: Simple Citrate anticoagulation protocol for Low Flux haemodialysis

Authors:

terina seow (123terina@gmail.com)
Eng Kuang Lim (lim.eng.kuang@alexandrahealth.com.sg)
Shun E Chen (chen.shune@alexandrahealth.com.sg)
Yang Gao (gg_yy_2001@yahoo.com.sg)
Min Er Liaw (liawminer@gmail.com)
Shimi Isaac (shimi_issac@yahoo.com)

Version: 3 Date: 28 Aug 2017

Author’s response to reviews:

3. Please confirm that consent was informed.

Yes, we have consent form in duplicate, 1 for record filing, 1 for patient’s retention

4. We note that an author is referred to as SYY in the authors contributions section, please confirm which author this is.

SYY is SEOW ying-ying, Terina, I have changed the author’s name.

In addition, author terina seow does not appear to be mentioned so please ensure that their contributions are listed.